Author's response to reviews

Title: Impact of air pollution on pulmonary function and respiratory symptoms in children. Longitudinal repeated-measures study.

Authors:

Benigno Linares (lisb700705@yahoo.com.mx)
Juan M Guizar (janguizar@prodigy.net.mx)
Norma Amador (norma.amador@imss.gob.mx)
Alfonso Garcia (algavela@hotmail.com)
Victor Miranda (vmsoberanis@gmail.com)
Jose R Perez (perezpad@servidor.unam.mx)
Rocio Chapela (rociochm@iner.gob.mx)

Version: 8 Date: 17 November 2010

Author's response to reviews:

BMC Series Editorial Production Team

The title is identical in the manuscript and in the submission system

We have changed the list of authors in the manuscript and now, they are written exactly as they are in the submission system, both in style and order.

References are cited in the text using consecutive numbers in square brackets.

We checked our manuscript for any typographical errors.

Sincerely yours

Juan M Guizar